Author's response to reviews

Title: Effect of micronutrient powder supplementation for two and four months on hemoglobin level of children 6-23 months old in a slum in Dhaka: a community based observational study

Authors:
Mustafa Mahfuz (mustafa@icddrb.org)
Mohammed Alam (mashraful@icddrb.org)
MMunirul Islam (mislam@icddrb.org)
Dinesh Mondal (din63d@icddrb.org)
MIqbal Hossain (ihossain@icddrb.org)
AM Ahmed (a.ahmed@uq.net.au)
Nuzhat Choudhury (nuzhat@icddrb.org)
Mohammad Raihan (raihan@icddrb.org)
Rashidul Haque (rhaque@icddrb.org)
Tahmeed Ahmed (tahmeed@icddrb.org)

Version: 2 Date: 21 Mar 2016

Author's response to reviews:
March 21, 2016

Editor
BMC Nutrition

Subject: Response to editors’ comments on NUTN-D-15-00058
Dear Dr Mockridge,

Thank you very much for provisionally accepting our manuscript titled "Effect of micronutrient powder supplementation for two and four months on hemoglobin level of children 6-23 months old in a slum in Dhaka: a community based observational study" (NUTN-D-15-00058). Below we respond to the editorial comments. We are expressing our sincere thanks for considering our manuscript.

EDITORIAL COMMENTS AND RESPONSES

1) Declarations: thank you for formatting the declaration section according to the new guidelines. Please could the Funding information be moved from the Acknowledgements into a section headed 'Funding'.

Response: Thank you very much for your suggestion. Now the revised manuscript has two separate sections for ‘Funding’ and ‘Acknowledgements’.

2) Figure 3: please remove the black background in this figure.

Response: We have removed the black background in figure 3.

3) Track changes: please ensure that all changes are approved and the track changes facility is turned off. A clean unmarked Word file is required.

Response: We have approved all changes in the manuscript and the track changes facility is now turned off.

Sincerely,

Mustafa Mahfuz